EO-1 Weekly Status Week of January 7 – January 13, 2010
Day of Year 007 - 013
Mission Day 3344 - 3350
Earth Observing-One (EO-1) General

Scheduled 122 Science Data Collection Events (DCEs) this week.
Another  Hyperspectral & HyspIRI Interest Group meeting was held on January 11 to address Center-wide interest in hyperspectral, thermal, and related technology issues for future space-based platforms, with 610 management support from Franco Einaudi and Shahid Habib.  There were 19 participants from different GSFC organizations attending the meeting that was organized by Dr. Betsy Middleton (614.4).  Bob Knox presented an overview of the HyspIRI concept and discussed the HyspIRI requirements traceability from the Decadal Survey to the Level 1 mission requirements.  The group recommended that the effect introduced by the HyspIRI larger pixel size (60 m) and wider swath size (145 km) be analyzed for sensitivity with respect to data products produced by algorithms when compared with Hyperion and AVIRIS results. 

INSTRUMENTS

All instruments operated nominally this week 

· Performed instrument calibration

· Conducted ALI internal calibration (Type II) on January 11 at 00:00z

EO-1 Spacecraft Subsystems

Command and Data Handling (C&DH)

Continued to experience problems playing back engineering data from the Solid-State-Recorders.  Engineering data is being received during real-time contacts and for all science imaging events. 

Technology Activities

In addition to providing ongoing science data collection, the EO-1 extended mission supports on-orbit testbed activities for advanced technology and hyperspectral research.  The status of various validation efforts is contained in the following paragraphs.

SensorWeb & Virtual Observatory Demonstrations
On Thursday, January 7, there was an AIST 2008 SensorWeb 3G collaborators teleconference.  

Participants:  Dan Mandl, Stu Frye, Vuong Ly, Rob Sohlberg, Jason Stanley, and Joe Young.

Notes from this teleconference that focused primarily on the HyspIRI IPM testbed are as follows:

1.  Installation of cFE on the Global Hawk testbed in February needs to be added to the tasks list.
2.  Rob Sohlberg is to add science tasks to the tasks list.

3.  Dan Mandl is to check on proposal submitted to the National Science Foundation by the Center for Ultra-Sustainable Platforms (CUSP) located at Northeastern University.
4.  Stu Frye recommended that the AIST task list be reconciled with the activity task list produced by Bogdan Oaida for the HyspIRI IPM.
On January 12, there was a Flood/Disease SensorWeb collaborators teleconference.  

Participants:  Dan Mandl, Stu Frye, Joerg Szarzynski, Lenny Roytman, Serhiy Skakun, Guido van Langenhove, Fritz Policelli, and Joe Young.

Notes from this teleconference are as follows: 

1.  There is a need for the Ukraine Space Research Institute to produce Envisat ASAR processed data for both the Caribbean and African flood pilot programs.
2.  Lenny Roytman has sent information on malaria that consists of bar graphs of occurrences for years 2001-2008.
3.  The agenda for the January workshop in Namibia was finalized with the addition of a possible wrap-up session on the morning of 30 Saturday.
4.  It was agreed that the January workshop should be a ”stepping stone” for the next UN-SPIDER meeting in Bonn, Germany possibly in May or June.   
5.  The World Bank will be represented at the workshop full time and is funding attendance of participants from neighboring countries.
6.  Lenny Roytman and Felix Kogan will be available during the workshop to be contacted by telephone when appropriate to discuss the health aspect
7.  Lenny is to talk with the Namibian Health Ministry contact person and discuss the workshop and urge him to attend.
8.  Guido van Langenhove is to send a second round of workshop invitations.
9.  Jan-Peter Mund and Joerg Szarzynski are to give an update on the short video either late this week or at next week’s teleconference.

On January 13 Bogdan Oaida, who is the JPL HyspIRI Mission Concept Manager, conducted another of an ongoing series of Webex teleconferences with GSFC team members where the status of the HyspIRI IPM testbed and coordination of associated activities between JPL and GSFC were discussed.  
Participants:  Bogdan Oaida, Carl Bruce, Steve Chien, Dan Mandl, Stu Frye, Jerry Hengemihle, and Joe Young.

Highlights of the teleconference are as follows:  
1.  There will be Planning Meeting on 22 February and a Team X meeting on 22-25 March.
2.  Jerry Hengemihle is to complete generating a functional block diagram of the IPM by January 22.
3.  Dan Mandl and Steve Chien are to compare their respective IPM operations concepts by 15 January.
4.  Jerry is to obtain the cost, mass, power, and volume and 3-D computer model for all IPM hardware by January 22.
5.  Dan is to produce an IPM development estimated cost and schedule by 22 January for Team X.
6.  Stu Frye was assigned an action to start list of ancillary data needed for algorithms. 

7.  Steve is to chair a session on Direct Broadcast for Decadal Survey Missions at a meeting in Colorado Springs on February 2-4.
8.  Next teleconference will be on 27 January.

Autonomous Sciencecraft Experiment (ASE)

The ASE controlled EO-1 all week. 

EO-1 MISSION OPERATIONS CENTER

Real-Time

· Continued nominal spacecraft state of health

· Continued analysis of recent increase in frequency of some instrument temperature limit violations—most notably the temperature of the ALI power module

· Obtaining number of images taken per day (and image types) in order to study effect on temperatures

· Plan to use data to develop any necessary operational constraints on imaging

Mission Planning

· Continued migration from Command Management System (CMS) software to ASIST SCP (Stored Command Processor) module

· Continued functional testing

· Found large file which is intended to only contain one week’s worth of logs

· Discovered file was constantly being appended, holding logs of all loads from inception to present

· Modified SCP code to prevent accumulation of dangerously large log files

· Found differences in expansion of RTS’s in text summary of command loads

· Discovered incorrect timestamps for SCP RTS expansions

· Working with developers to correct  issue

Flight Dynamics
· Collaborating with system administrators at JPL, USGS, and other collaborators to finalize transition of flight dynamics file deliveries to new computer

Trending

· Monitored the newly installed script on the trending system

System Administration
· Continuing benchmark of ASPEN computers which will replace old mission planning system

· Supporting C&S audit scheduled for February

· Determining responsibility for legacy systems in FSW lab that are on private network, which were administered by outside entities earlier in mission

· Collaborating with system administrators at other agencies to finalize transition of flight dynamics file deliveries to new computer

GROUND AND SPACE NETWORK

Station Downtimes

· Received notification that HGS has returned to nominal status

Operational Discrepancies
· Observed no operational discrepancies
UPCOMING EVENTS

· In January 2010, plan to institute ASE software patch onboard spacecraft to correct for the leap second issue, the WARP Mongoose V single bit error issue, and the temperature log size issue. 

Imagery Status

Scenes and Engineering Cals planned for week of January 7 – January 13, 2009             122
Total scenes and engineering calibrations planned for entire mission – approximately 50,825
Total Scenes:  ALI scenes in the Level 0 archive              46,453             (as of January 13, 2010)

                         Hyperion scenes in the Level 0 archive    46,205             
Publications and Presentations Status 

404 publications 

284 external presentations

53 articles and press releases
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